Author's response to reviews

Title: Regional disparities in interferon therapy for chronic hepatitis C in Japan: A nationwide, retrospective cohort study

Authors:

Naohiko Masaki (nmasaki0808@gmail.com)
Yoko Yamagiwa (lbyamagiwa@hospk.ncgm.go.jp)
Takuro Shimbo (tshimbo@ohta-hp.or.jp)
Kazumoto Murata (dmurata@hospk.ncgm.go.jp)
Masaaki Korenaga (dmkorenaga@hospk.ncgm.go.jp)
Tatsuya Kanto (kantot@hospk.ncgm.go.jp)
Masashi Mizokami (mizokami0810@gmail.com)

Version: 4
Date: 23 May 2015

Author's response to reviews: see over
June 23, 2015.

Prof. Carl Latkin
Editor
BMC Public Health

Re: Submission of the final version of the manuscript (MS: 1684795701394958) entitled, “Regional disparities in interferon therapy for chronic hepatitis C in Japan: A nationwide retrospective cohort study”

Dear Prof. Latkin,

Please find attached the above final version of our manuscript submitted for publication as an original research article in BMC Public Health. We believe that the findings reported in this study are relevant to the scope of your journal and will be of interest to its readership.

We would like to respond point by point to the request of further editorial revisions.

1) The map depicted in Figure 1 is our own original for this manuscript.
2) According to your request, I added the e-mail addresses for the coauthors in the title page of the manuscript.
3) I would ensure that our revised manuscript conforms to your journal style.

We hope that our responses to your comments are sufficient to allow publication of our paper in BMC Public Health.

We look forward to hearing from you at your earliest convenience.

Yours sincerely,

Naohiko Masaki, MD, PhD.
The Research Center for Hepatitis and Immunology,
National Center for Global Health and Medicine,
1-7-1 Kohnodai, Ichikawa, Chiba 272-8516, Japan
Tel: +81-47-375-4753; Fax: +81-47-375-4786
E-mail: nmasaki0808@gmail.com